Author’s response to reviews

Title: Premature ovarian insufficiency as a variable feature of blepharophimosis, ptosis, and epicanthus inversus syndrome associated with p.Leu75Phe FOXL2 mutation: a case report.

Authors:
Barbara Grzechocińska (bgrzechocinska@wum.edu.pl)
Damian Warzecha (damwarzecha@gmail.com)
Maria Wypchło (maria.wypchlo@gmail.com)
Rafal Ploski (rploski@wp.pl)
Mirosław Wielgoś (miroslaw.wielgos@wum.edu.pl)

Version: 3 Date: 10 Jun 2019

Author’s response to reviews:

BMC Medical Genetics
Dear Editor-in-Chief,

We enclose a revised version of the paper entitled “Premature ovarian insufficiency as a variable feature of blepharophimosis, ptosis, and epicanthus inversus syndrome associated with p.(Leu75Phe) FOXL2 mutation: a case report”, which was previously submitted for publication in the BMC Medical Genetics (MGTC-D-18-00417).

We are very grateful for giving us the opportunity to revise the manuscript. We would also like to thank all the Reviewers for their time and all the valuable remarks, which allowed us to improve our paper. We tried to answer all the questions and correct whatever was suggested. All the changes in the manuscript are marked in yellow.

Point-by-point response to all the remarks:

Response to the Editor:
1. The manuscript has been checked by native English speaker
2. All previous suggestions has been already introduced
3. High resolution version of Figure 3 has been introduced.

Responses to the Reviewers:

Ad. Reviewer 1 (Petra Liskova):

General comments: the manuscript has been improved by native English speaker

Minor corrections:

1. Changed.
2. Changed.
3. Changed.
4. Changed.
5. Changed.
6. Changed.
7. Changed.
8. Changed.
9. Changed.
10. Changed.
11. Changed.

All authors have read the revised version of the manuscript and approved its contents.

Thank you very much for considering our manuscript for publication.

Yours sincerely,

Damian Warzecha MD,

Corresponding Author